Author's response to reviews

Title: Boys are more stunted than girls in Sub-Saharan Africa: a meta-analysis of 16 demographic and health surveys

Authors:

Henry Wamani (wamanih@yahoo.com)
Anne A Nordrehaug (Anne.Aastrom@cih.uib.no)
Stefan Peterson (Stefan.Peterson@phs.ki.se)
James K Tumwine (jtumwine@imul.com)
Thorkild Tylleskar (Thorkild.Tylleskar@cih.uib.no)

Version: 4 Date: 5 April 2007

Author's response to reviews:

The following editorial changes have been made following your advice:
1. The sequencing of list of authors has been written as it appears in the submission system
2. The text "Authors has been deleted from the title page
3. The abstract has been inserted to the manuscript and key words removed
4. The term "et al" has been removed from references
5. The term "competing interests" has been used and appropriate wording used
6. The column headings for table 3 are no longer repeating